Author’s response to reviews

Title: Surgery versus Plaster in the Treatment of Distal Radius Fractures

Authors:

Monique MJ Walenkamp (m.m.walenkamp@amc.nl)
J Carel Goslings (j.c.goslings@amc.nl)
Annechien Beumer (beumera@ampa.nl)
Robert Haverlag (r.haverlag@olvg.nl)
Peter A Leenhouts (leenhouts.p@zaansmc.nl)
Egbert JJM Verleisdonk (ejverlei@diakhuis.nl)
Ronald SL Liem (Ronald.liem@ghz.nl)
Jan Bernard Sintenie (jbs@elkerliek.nl)
Maarten WGA Bronkhorst (mbronkhorst@bronovo.nl)
Jasper Winkelhagen (j.winkelhagen@westfriesgasthuis.nl)
Niels WL Schep (n.w.schep@amc.nl)

Version: 2 Date: 23 April 2013

Author’s response to reviews:

Amsterdam, April 22, 2013

Dear members of the editor of BMC,

Thank you very much for considering our manuscript for publication in your journal. We have made and tracked the changes you requested. Below you will find a point-by-point description of the alterations.

• On page 6, the section on the principal investigator and contact details was deleted to adhere to the requested order of sections.
• On page 18 and 19 the authors’ contributions are now listed.
• On page 19 an acknowledgements section was added.

With kind regards,

On behalf of the VIPER study group,

Monique Walenkamp